Author’s response to reviews

Title: Assessing the role of criminality in neighbourhood safety feelings and self-reported health. Results from a cross-sectional study in a Dutch municipality

Authors:

Polina Putrik (polina.putrik@maastrichtuniversity.nl)
Ludovic van Amelsvoort (lgpm.vanamelsvoort@maastrichtuniversity.nl)
Suhreta Mujakovic (Suhreta.Mujakovic@ggdzl.nl)
Anton Kunst (a.kunst@amc.uva.nl)
Hans van Oers (ja.v.oers@minvws.nl)
IJmert Kant (ij.kant@maastrichtuniversity.nl)
Maria Jansen (Maria.Jansen@ggdzl.nl)
Nanne De Vries (n.devries@maastrichtuniversity.nl)

Version: 1 Date: 14 Jun 2019

Author’s response to reviews:

Dear Editor,

Thank you for your letter. In response to the last query, please be advised that we have uploaded the new version of the manuscript. Figures have been taken out from the manuscript during the preceding review round, therefore we have now also removed the references to figures from the text (it was previously overlooked).

Best regards,

Polina Putrik on behalf of the authors